Author’s response to reviews

Title: THE ROLE OF THE GENERAL PRACTITIONER IN MULTIDISCIPLINARY TEAMS: A QUALITATIVE STUDY IN ELDERLY CARE

Authors:

Sietske Grol (Sietske.Grol@radboudumc.nl)
Gerard Molleman (gmolleman@ggdgelderlandzuid.nl)
Anne Kuijpers (anne.kuijpers@radboudumc.nl)
Rob van der Sande (Rob.vanderSande@han.nl)
Gerdine Fransen (GFransen@ggdgelderlandzuid.nl)
Willem Assendelft (pim.assendelft@radboudumc.nl)
Henk Schers (henk.schers@radboudumc.nl)

Version: 2 Date: 22 Feb 2018

Author’s response to reviews:

Dear Janelle Coore,

1) Title Page

Please include a title page as the first page of your manuscript where you list the full names, institutional addresses and email addresses for all authors. The corresponding author should be indicated.

We included a title page.

2) Declarations Heading
Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Ethics approval and consent to participate
Consent for publication
Availability of data and material
Competing interests
Funding
Authors' contributions
Acknowledgements

All declarations are included.

3) Availability of Data and Materials

Please edit the ADM heading to read 'Availability of Data and Materials'.

The heading has been edited.

4) Appendix Files

Please remove the appendix files from the main manuscript and upload them as additional files.
The appendix files will be uploaded as additional files.

5) Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript had been cleaned from tracked changes etc..

Kind regards,

Sietske Grol